Author's response to reviews

Title: A multi-disciplinary education process related to the discharging of children from hospital when the child has been diagnosed with type 1 diabetes - a qualitative study

Authors:

Lisbeth Jönsson (lisbeth.jonsson@med.lu.se)
Inger Hallström (inger.hallstrom@med.lu.se)
Anita Lundqvist (anita.lundqvist@med.lu.se)

Version: 4 Date: 24 February 2010

Author's response to reviews:

Dear editor of BMC Pediatric

We herewith submit the third version of the manuscript MS 1571320526324565 “A multi-disciplinary education process related to the discharging of children from hospital when the child has been diagnosed with type 1 diabetes - a qualitative study” for publication in BMC Pediatric. The purpose of this study was to seek a deeper understanding of how the diabetes team’s parent / child education process works, from admission to discharge, among families with a child newly diagnosed with type 1 diabetes.

We are grateful for all comments from referee and have revised the manuscript in all points recommended from both of the referees and a point-by-point response to the concerns is attached. The language is once again extensively edited as also was requested.

Now we look forward to hear from you and hope for the best. However, it is somewhat surprising that despite the fact that we revised the manuscript in all respects wanted the first time, the level of interest has been changed from “An article for importance for its field” to “An article of limited interest”.

The undersigned authors transfer all copyright ownership of the manuscript to BMC Pediatric in the event that the manuscript is published. The manuscript is original, is not under consideration by another journal, and has not been published previously. There is no conflict of interest.

Yours sincerely

Lisbeth Jönsson Inger Hallström Anita Lundqvist
RN, RSCN, PhD student RN, RSCN, Professor RN, RSCN, PhD